Author's response to reviews

Title: Coexistence of pernicious anemia and prostate cancer - 'an experiment of nature' involving vitamin B12 modulation of prostate cancer growth and metabolism: a case report

Authors:

   Glenn Tisman (glennmd@gmail.com)
   Seth Kutik (glennmd@gmail.com)
   Christa Rainville (glennmd@gmail.com)

Version: 7 Date: 20 November 2009

Author's response to reviews:

We have made all suggested changes listed below:

Formatting changes made:

----------------------------

Major revisions (we require the authors to make these changes)

Figure citations: do not use ranges when citing figures, please cite each individually.

Author's contributions: please indicate that all authors have 'read and approved the final version of the manuscript'.

References: all references must be cited in consecutive numerical order. Currently, references 11 and 12 are cited before reference 10, please amend this.

Minor revisions (we can make these changes for the authors, however it will speed publication of your article if you make these revisions along with the major revisions above)

Title page: please remove the word 'address' and the received and accepted sections.